Author's response to reviews

Title: Expression and biological-clinical significance of hTR, hTERT and CKS2 in washing fluids of patients with bladder cancer.

Authors:

Letizia Mezzasoma (lmezzaso@unipg.it)
Cinzia Antognelli (cinzia.antognelli@unipg.it)
Chiara Del Buono (chiaradelbuono@libero.it)
Fabrizio Stracci (fabs@unipg.it)
Emanuele Cottini (emanuelecottini@libero.it)
Giovanni Cochetti (giovannicochetti@libero.it)
Vincenzo N Talesa (talesa@unipg.it)
Ettore Mearini (mearini@unipg.it)

Version: 4 Date: 27 September 2010

Author's response to reviews: see over
Dear Editor,

I am submitting the revised version of manuscript entitled: “Expression and biological-clinical significance of hTR, hTERT and CKS2 in washing fluids of patients with bladder cancer” by Letizia Mezzasoma et al.

Formatting changes have now been made. In particular:

1) the list of authors in the manuscript has been written exactly as they are in the submission system, both in style and order.

2) all the highlighted parts have been removed from the manuscript.

3) “Material and methods” section has been renamed just “Methods”.

4) the image files do not include the title anymore and have been cropped as closely as possible to minimize white space around the image.

5) the manuscript has been checked for any typographical errors and any final corrections or revisions have been made.

Yours sincerely

Letizia Mezzasoma

(On the behalf of all authors)